////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-Collator]]++Collator++::
** [[painless-api-reference-Collator-CANONICAL_DECOMPOSITION]]static int link:{java8-javadoc}/java/text/Collator.html#CANONICAL_DECOMPOSITION[CANONICAL_DECOMPOSITION] (link:{java9-javadoc}/java/text/Collator.html#CANONICAL_DECOMPOSITION[java 9])
** [[painless-api-reference-Collator-FULL_DECOMPOSITION]]static int link:{java8-javadoc}/java/text/Collator.html#FULL_DECOMPOSITION[FULL_DECOMPOSITION] (link:{java9-javadoc}/java/text/Collator.html#FULL_DECOMPOSITION[java 9])
** [[painless-api-reference-Collator-IDENTICAL]]static int link:{java8-javadoc}/java/text/Collator.html#IDENTICAL[IDENTICAL] (link:{java9-javadoc}/java/text/Collator.html#IDENTICAL[java 9])
** [[painless-api-reference-Collator-NO_DECOMPOSITION]]static int link:{java8-javadoc}/java/text/Collator.html#NO_DECOMPOSITION[NO_DECOMPOSITION] (link:{java9-javadoc}/java/text/Collator.html#NO_DECOMPOSITION[java 9])
** [[painless-api-reference-Collator-PRIMARY]]static int link:{java8-javadoc}/java/text/Collator.html#PRIMARY[PRIMARY] (link:{java9-javadoc}/java/text/Collator.html#PRIMARY[java 9])
** [[painless-api-reference-Collator-SECONDARY]]static int link:{java8-javadoc}/java/text/Collator.html#SECONDARY[SECONDARY] (link:{java9-javadoc}/java/text/Collator.html#SECONDARY[java 9])
** [[painless-api-reference-Collator-TERTIARY]]static int link:{java8-javadoc}/java/text/Collator.html#TERTIARY[TERTIARY] (link:{java9-javadoc}/java/text/Collator.html#TERTIARY[java 9])
* ++[[painless-api-reference-Collator-getAvailableLocales-0]]static <<painless-api-reference-Locale,Locale>>[] link:{java8-javadoc}/java/text/Collator.html#getAvailableLocales%2D%2D[getAvailableLocales]()++ (link:{java9-javadoc}/java/text/Collator.html#getAvailableLocales%2D%2D[java 9])
* ++[[painless-api-reference-Collator-getInstance-0]]static <<painless-api-reference-Collator,Collator>> link:{java8-javadoc}/java/text/Collator.html#getInstance%2D%2D[getInstance]()++ (link:{java9-javadoc}/java/text/Collator.html#getInstance%2D%2D[java 9])
* ++[[painless-api-reference-Collator-getInstance-1]]static <<painless-api-reference-Collator,Collator>> link:{java8-javadoc}/java/text/Collator.html#getInstance%2Djava.util.Locale%2D[getInstance](<<painless-api-reference-Locale,Locale>>)++ (link:{java9-javadoc}/java/text/Collator.html#getInstance%2Djava.util.Locale%2D[java 9])
* ++[[painless-api-reference-Collator-clone-0]]def link:{java8-javadoc}/java/text/Collator.html#clone%2D%2D[clone]()++ (link:{java9-javadoc}/java/text/Collator.html#clone%2D%2D[java 9])
* ++[[painless-api-reference-Collator-equals-2]]boolean link:{java8-javadoc}/java/text/Collator.html#equals%2Djava.lang.String%2Djava.lang.String%2D[equals](<<painless-api-reference-String,String>>, <<painless-api-reference-String,String>>)++ (link:{java9-javadoc}/java/text/Collator.html#equals%2Djava.lang.String%2Djava.lang.String%2D[java 9])
* ++[[painless-api-reference-Collator-getCollationKey-1]]<<painless-api-reference-CollationKey,CollationKey>> link:{java8-javadoc}/java/text/Collator.html#getCollationKey%2Djava.lang.String%2D[getCollationKey](<<painless-api-reference-String,String>>)++ (link:{java9-javadoc}/java/text/Collator.html#getCollationKey%2Djava.lang.String%2D[java 9])
* ++[[painless-api-reference-Collator-getDecomposition-0]]int link:{java8-javadoc}/java/text/Collator.html#getDecomposition%2D%2D[getDecomposition]()++ (link:{java9-javadoc}/java/text/Collator.html#getDecomposition%2D%2D[java 9])
* ++[[painless-api-reference-Collator-getStrength-0]]int link:{java8-javadoc}/java/text/Collator.html#getStrength%2D%2D[getStrength]()++ (link:{java9-javadoc}/java/text/Collator.html#getStrength%2D%2D[java 9])
* ++[[painless-api-reference-Collator-setDecomposition-1]]void link:{java8-javadoc}/java/text/Collator.html#setDecomposition%2Dint%2D[setDecomposition](int)++ (link:{java9-javadoc}/java/text/Collator.html#setDecomposition%2Dint%2D[java 9])
* ++[[painless-api-reference-Collator-setStrength-1]]void link:{java8-javadoc}/java/text/Collator.html#setStrength%2Dint%2D[setStrength](int)++ (link:{java9-javadoc}/java/text/Collator.html#setStrength%2Dint%2D[java 9])
* Inherits methods from ++<<painless-api-reference-Comparator,Comparator>>++, ++<<painless-api-reference-Object,Object>>++
